Author’s response to reviews

Title: Statistical software applications used in health services research: analysis of published studies in the U.S.

Authors:

Allard E Dembe (adembe@cph.osu.edu)
Jamie S Partridge (jpartridge@cph.osu.edu)
Laurel C Geist (geist.32@osu.edu)

Version: 4 Date: 21 September 2011

Author’s response to reviews:

September 20, 2011

Cover Letter

I am submitting the enclosed revised manuscript entitled, “Statistical software applications used in health services research: analysis of published studies” to BMC: Health Services Research. All the changes that were requested in your email of September 13, 2011 have been made including:

1. Making sure the abstract in the manuscript is identical to the abstract in the submission system.

2. In the authors’ contribution section, we added a sentence to indicate that all authors have read and approved the final manuscript.

3. We have removed additional files from the manuscript.

4. We have removed job titles from the author list.

5. We have included country identification in the author list.

6. We have removed the title from the abstract page.

7. We have again proofread the manuscript to ensure that there are no typographical errors.

Please let us know if there is anything else that you need. Thank you.

Very truly yours,

Allard E. Dembe, ScD, Chair
Division of Health Services Management and Policy
The Ohio State University College of Public Health